Author's response to reviews

Title: Near-highway pollutants in motor vehicle exhaust: A review of epidemiologic evidence of cardiac and pulmonary health risks

Authors:

Doug Brugge (dbrugge@aol.com)
John L Durant (john.durant@tufts.edu)
Christine Rioux (Christine.rioux@tufts.edu)

Version: 6 Date: 6 August 2007

Author's response to reviews:

Figure was cropped again.

Headings are now all in sentence case.